Effects of environmental temperature on the effectiveness of microbially induced carbonate precipitation

Yuze Wang¹, Yong Wang¹, and Charalampos Konstantinou²

¹Department of Ocean Science and Engineering, Southern University of Science and Technology

²Department of Civil and Environmental engineering, University of Cyprus

November 24, 2022

Abstract

Microbial-induced calcium carbonate precipitation (MICP) is an innovative technique used for soil improvement, for controlled reduction of permeability in porous media or immobilization of soil contaminants. The application of MICP in the field is influenced by the environmental factors. In the present study, the main purpose is to explore the effectiveness of MICP in treating porous media at different environmental temperatures and reveal the underlying mechanisms. The microstructure characteristics were investigated via SEM imaging, EDS and XRD analyses and consolidated drained triaxial compression tests were performed to examine the performance of MICP-treated samples. Results indicate that the shear strength depends heavily on the treatment temperature, which was mainly due to the different content, size and distribution of CaCO₃ in samples at different conditions. The observations of pore-scale characteristics revealed that low temperature (4) and high temperature (50) produced less CaCO₃ precipitation, resulted in smaller carbonate crystals precipitation and thus lower strength. In contrast, samples treated at room temperature and 35 show more CaCO₃ precipitation and greater strength. The crystal forms, though, were not influenced by the temperature. The climate conditions are a very important parameter that needs to be tuned specifically for the purposes of each MICP application (whether controlled alteration of permeability or for soil stabilization). However, in most MICP field applications, temperature is nearly impossible to control, and in such conditions where bacterial activity is reduced, the alteration of the MICP recipe is required, and specifically the number of bacterial solution injections are worth to be considered.

Hosted file

essoar.10510835.1.docx available at https://authorea.com/users/549722/articles/603696effects-of-environmental-temperature-on-the-effectiveness-of-microbially-inducedcarbonate-precipitation

Hosted file

tables.docx available at https://authorea.com/users/549722/articles/603696-effectsof-environmental-temperature-on-the-effectiveness-of-microbially-induced-carbonateprecipitation

Effects of environmental temperature on the effectiveness of microbially induced carbonate precipitation

Yuze Wang^{a,b*}, Yong Wang^a, Charalampos Konstantinou^c

a. Department of Ocean Science and Engineering, Southern University of Science and Technology, Shenzhen, 518055, China

b. Southern Marine Science and Engineering Guangdong Laboratory (Guangzhou), Shenzhen, 518055, China

c. Department of Civil and Environmental Engineering, University of Cyprus, Nicosia, Cyprus

Corresponding author: Yuze Wang; wangyz@sustech.edu.cn

Abstract

Microbial-induced calcium carbonate precipitation (MICP) is an innovative technique used for soil improvement, for controlled reduction of permeability in porous media or immobilization of soil contaminants. The application of MICP in the field is influenced by the environmental factors. In the present study, the main purpose is to explore the effectiveness of MICP in treating porous media at different environmental temperatures and reveal the underlying mechanisms. The microstructure characteristics were investigated via SEM imaging, EDS and XRD analyses and consolidated drained triaxial compression tests were performed to examine the performance of MICP-treated samples. Results indicate that the shear strength depends heavily on the treatment temperature, which was mainly due to the different content, size and distribution of $CaCO_3$ in samples at different conditions. The observations of pore-scale characteristics revealed that low temperature (4°C) and high temperature (50°C) produced less CaCO₃ precipitation, resulted in smaller carbonate crystals precipitation and thus lower strength. In contrast, samples treated at room temperature and 35 [°]C show more CaCO₃ precipitation and greater strength. The crystal forms, though, were not influenced by the temperature. The climate conditions are a very important parameter that needs to be tuned specifically for the purposes of each MICP application (whether controlled alteration of permeability or for soil stabilization). However, in most MICP field applications, temperature is nearly impossible to control, and in such conditions where bacterial activity is reduced, the alteration of the MICP recipe is required, and specifically the number of bacterial solution injections are worth to be considered.

Plain Language Summary

We studied the effects of MICP treatment of granular material under various temperatures, that are representative of the cold, tropical, semi-arid (35°C) and arid regions. Results show that the strength enhancement is affected by the

climate conditions since different precipitation patterns of calcium carbonate crystals were observed. Temperature is a very important parameter that needs to be taken into account when designing MICP protocols for field applications whether these are soil permeability control, or soil stabilization.

Keywords: environmental temperature; climate conditions; urease activity; MICP performance; shear strength

Key points:

- Microbially induced carbonate precipitation was applied in granular material under different temperature conditions.
- Microscale analysis showed that the precipitation patterns vary significantly depending on the temperature.
- The strength of the resulting products follows the same trend as the precipitation patterns.

1 Introduction

Microbial-induced calcium carbonate precipitation (MICP) is a biochemical process in which microorganisms induce an enzyme-catalyzed process that increases the supersaturation ratio of calcium carbonate and induces its precipitation in an aqueous environment (Mitchell and Santamarina 2005; DeJong et al. 2006). When the process is applied to soil pore water, precipitated calcium carbonate minerals form at the pore throats of the soil matrix and on the surface of soil particles, increasing soil binding and soil surface roughness, thereby increasing soil strength and stiffness. Therefore, MICP has been widely studied for applications including soil stabilisation (DeJong et al., 2010, 2011, 2013; van Paassen et al. 2010; Umar et al. 2016; Gowthaman et al. 2019; Naveed et al. 2019) and the generation of artificially cemented sandstones from sand for hydraulic fracturing experiments (Konstantinou et al. 2021). In addition, the amount of precipitated calcium carbonate minerals in the soil pores can be controlled to not fully occupy the whole soil pores, thereby maintaining a relatively high soil permeability or providing more control on permeability alteration compared to other cementation methods. Therefore, MICP can also be potentially used for liquefaction (Montoya et al. 2013; Xiao et al. 2018; Darby et al. 2019) and erosion control (Jiang et al. 2017; Chek et al. 2021). Moreover, MICP can also seal cracks of soil or geo-structures for applications such as CO_2 storage, and concrete remediation (Minto et al., 2016; Tobler et al. 2018; Weinhardt et al. 2021). Furthermore, carbonate can co-precipitate with heavy metal ions, so it can solidify heavy metals in soil and water (Warren et al. 2001; Fujita et al. 2004; Li et al. 2013; Mugwar and Harbottle 2016; Torres-Aravena et al. 2018; Jiang et al. 2019; Mwandira et al. 2017).

Ureolysis-driven MICP is the most widely studied MICP process for soil stabilization, among the others such as denitrification and sulfate reduction, due to its ease of control, high chemical conversion efficiency and short amount of time required (Dhami et al. 2013, Castro-Alonso et al. 2019). In this process, the ureolysis bacteria cells were introduced into soil matrix by either bio-augmentation (Gomez et al. 2017, 2019) or bio-stimulation (Wang et al. 2020; Graddy et al. 2021) followed with the injections of multiple times cementation solution which include CaCl₂, urea and nutrient broth. Ureolysis bacteria express a urease enzyme which catalyses the hydrolysis of urea (*Equation 1*); the addition of calcium (Ca²⁺) to this system induces the precipitation of calcium carbonate (CaCO₃) as the CO₃²⁻ ions produced by the hydrolysis of urea react with the supplied Ca²⁺ (*Equation 2*) (van Paassen et al., 2010; Al Qabany and Soga, 2013); the addition of nutrient broth to this system maintains the bacterial activity over the precipitation period (Al Qabany and Soga, 2013):

(1)

(2)

MICP rate and efficiency are directly affected by parameters such as bacterial activity (Wang et al. 2021; Konstantinou et al. 2021; Omoregie et al. 2017), concentration of cementation solution (De Muynck et al. 2010; Al Qabany and Soga 2013), urea to calcium chloride ratio (Okwadha and Li 2010; Mahawish et al. 2018a), bacterial and chemical injection methods (Cheng and Cord-Ruwisch 2012, 2014; Cheng and Shahin 2016; Martinez et al. 2013; Wu et al. 2019) and the frequency of injections or retention period (Al Qabany et al. 2013; Qin et al. 2016; Lee Lee et al. 2013). In addition, soil particles properties including grain size and size (Mahawish et al. 2018b; Zamani et al. 2019) and particle size distribution (Xiao et al. 2021) were shown to affect the effective of MICP for soil strengthening. Further, environmental factors, including temperature (Peng et al. 2019; Sun et al. 2019) salinity and aqueous conditions (Mortensen et al. 2011; Soon et al. 2014) have also arisen research attention.

For MICP to be able to be used in large scale field applications, the considerations of environmental factors on MICP efficiency and optimization are essential. Among these environmental factors, climate is a crucial one. The subsurface temperature in different areas normally falls within the range between 10 and 30°C (Hu and Feng 2003; Chow et al. 2011; Kumarathunge et al. 2013), whilst it could even become lower than 10 °C or higher than 50°C for some cold or arid regions, respectively (Gao et al. 2007; Huang et al. 2020). Such extreme drought events are much more frequent in the recent years because of climate change which introduces wormer summers and colder winters. Any real-field MICP implementation strategy (for controlling permeability, soil stabilization or contaminant transport control) needs to consider the soil temperature. Studies have shown that temperature influences the $CaCO_3$ precipitation rate, crystal properties, thus soil permeability (Cheng et al. 2017; Peng et al. 2019) and unconfined compressive strength (UCS) (Cheng et al. 2017), Recently, the effects of temperature on the MICP processes and properties were studied by Wang et al. (2022). Despite the studies conducted, the effect of temperature

on the properties of soils treated by MICP at different temperatures were studied in a systematic way. Since triaxial shear testing parameters are important parameters used to define yielding and failure, the evaluation of soil strength enhancement obtained at field temperature properties by triaxial shear testing is essential for all listed MICP field applications.

This research investigates the microstructure characteristics at the pore scale (amount, size, distribution and form of carbonate crystals) and the mechanical response of the bio-cemented specimens generated at different temperatures simulating the field environment including cold (4°C), tropical (20°C), semi-arid (35°C), and arid regions (50°C). The aim is to investigate the effect of treatment temperature on microstructure and mechanical behavior of MICP-treated soils, especially the shear behavior, with the main objective to explore the feasibility of field application of MICP-treated soil. A series of triaxial compression tests were conducted accompanied with SEM imaging, EDX and XRD analyses to estimate fulfill the objectives of this study. Triaxial testing was used to derive cohesion and friction angle which are the ultimate parameters used to define yielding and failure. The mechanical response was correlated with the microstructure characteristics since the calcium carbonate precipitation rate, crystals size and location within the grains affect the strength of the resulting MICP products.

2 Materials and methods

2.1 Sand properties and specimen preparation

Fine CHINA ISO standard silica sand was used in this study, and its main properties are summarized in Table 1. The particle size distribution of the sand is demonstrated in Fig. 1. The mean grain size, D_{10} , and coefficient of uniformity, $C_{\rm u}~(=D_{60}/D_{10})$, are 0.125 mm and 5.6, respectively, as shown in Fig. 1. The sand is classified as poorly graded according to the Unified Soil Classification System (ASTM, 2017).

A split acrylic cylindrical mold with an inner diameter of 38 mm and a height of 80 mm was used to prepare the specimens (Fig. 2). The weight of dry sand was calculated based on a targeted relative density (RD) of 50%. Sand was poured into the columns in 3 portions via dry pluviation method. After the sand column was prepared, deionized water was flushed in the column to expel air from the columns. This step was repeated several times until the flow rate was stabilized and the specimen became fully saturated. Then, the drain valve was shut such that the sand column remained saturated until the biotreatment process began.

2.2 Biological and chemical treatment procedure

S. pasteurii (CGMCC1.3687) was used in this study as it is a urease-active strain and its urease activity is well demonstrated over many other alternative

ureolytic bacteria. The bacterial strain was cultivated in a NH₄-YE liquid media (ATCC 1366) for about 24 h at 30°C in a shaking incubator at 200 rpm/min to obtain an optical density at 600 nm (OD₆₀₀) of 1.0. The tested bacterial activity of the bacterial suspension at (OD₆₀₀) of 1.0 was approximately 40 mM/h. The cementation solution consisted of 0.75 M of urea, 0.5 M of calcium chloride, and 3 g/L of nutrient broth. The grade level of all chemical reagents was analytical grade.

A gravity filtration method was used for MICP treatment process (injection from top to bottom via gravity). Before MICP treatment, the pore volume (PV) of sand columns was calculated using the targeted relative density, so that the injected volume of bacterial suspension (BS) and cementation solution (CS) during each cycle could be determined. The prepared saturated sand column specimen was treated by the two-phase injection method, where certain volumes of BS were injected into the soil matrix, followed by a bacterial settling period to allow bacterial cells to attach to soil particles (Phase 1). Then, the next step was the injections of CS at a designed injection interval (Phase II) to allow MICP to occur. The parameters of MICP treatment are shown in Table 2.

In total, six tests were conducted to assess the effects of temperature on the resulting microstructure characteristics and strength of the MICP products. In tests 1-4, the bacterial settling time and the time interval across the cementation solution injections was 24 hours. Since it was found that the bacterial urease activity decreases rapidly at 50°C (Wang et al. 2022), the number of BS injections in tests 5 and 6 for 50°C case was increased to 3, and 6, respectively, and the time between BS and CS injection was shortened at 2 hours instead of 24 hours in these two tests. In all tests, in either BS or CS injections, 1.0 PV of BS or CS was injected each time, and the total number of CS injections was 6. Before MICP treatment, the pore volume (PV) of sand columns was calculated using the targeted relative density, so that the injected volume of bacterial suspension (BS) and cementation solution (CS) during each cycle could be determined. The samples were placed at the designed temperature during all retention times of the experiment. The four temperatures, 4°C, 20°C. 35°C and 50°C were achieved by placing the setup in a refrigerator, at room temperature, and in two ovens.

2.3 Triaxial tests

After completing the biological treatment process the specimens were carefully extracted from the molds for conducting consolidated drained (CD) triaxial tests (ASTM 2020) to determine of strength and stress-strain relationships of a cylindrical specimen of MICP-treated specimens at these temperatures. Specimens were firstly saturated with vacuum saturation for about 24 h and then placed in the triaxial apparatus. Prior to consolidation, specimens were saturated by hydraulic saturation for 12 h at a confining pressure of 20 kPa and then back-pressure saturation was sequentially applied until the B-value exceeded 0.95. For each experimental condition, four samples were prepared, and the triaxial spec-

imens were consolidated at an effective confining pressure of 100, 200, 300, and 400 kPa, respectively. Following consolidation, specimens were sheared under drained conditions with a constant displacement loading rate of 0.1 mm/min.

2.4 $CaCO_3$ content measurement and chemical efficiency calculation

The ASTM Method (ASTM 2014) was adopted for the determination of the $CaCO_3$ content of biotreated sand. A calibration curve was first developed by adding various amounts of calcite $(CaCO_3)$ with hydrochloric acid and measuring the CO_2 pressure. As shown in Eq. (3), a linear relationship between CO_2 pressure and $CaCO_3$ content was developed. After the triaxial tests were completed, the specimens were carefully removed from the membrane and 15 to 25 g of specimens were collected for every 15 mm along the sand column height. The samples were oven-dried at 105°C for at least 24 h (ASTM 2014). Once removed from the oven, the specimens were grounded, weighed, and placed into the $CaCO_3$ measurement chamber. 30 ml of 3 M hydrochloric acid was added to a container and then placed into the chamber without contacting the soil specimens. The CaCO₃ measurement camber was completely sealed and shaken gently to facilitate the reaction between CaCO₃ and hydrochloric acid. The generatured CO_2 from the reaction between $CaCO_3$ and hydrochloric acid increase the pressure of the camber and was measured by a pressure gauge inserted on the top of the camber. When the pressure value indictated by the gauge no longer changed, the reading was recorded and the amount of CaCO₃ was calculated utilising Eq. (3).

(3)

The chemical conversion efficiency of MICP was then calculated, which is defined as the ratio of the precipitated mass of $CaCO_3$ in the sand to the calculated mass of $CaCO_3$ from cementation solutions (Al Qabany et al. 2012; Wang 2018):

(4)

where, is the measured $CaCO_3$ content, is the concentration of $CaCl_2$ in the cementation solutions V is total the volume of cementation solution injected into samples is the molar mass of $CaCO_3$ (100 g/mol); is the dry mass of sand used to prepare sample columns. It should be noted that the calculated chemical conversion efficiency may be lower than the actual value, because the $CaCO_3$ precipitated on the mold surface and in the filter layers at the top and bottom of the sample was not taken into account when calculated according to Eq (4).

2.5 Scanning electron microscopy (SEM) imaging, EDS and XRD analyses

Samples of biocemented sand treated at different temperatures were oven-dried at 105°C and prepared for Scanning electron microscopy (SEM) imaging to observe the microscale properties of the $CaCO_3$ crystals once formed after the

MICP treatment. Scanning electron microscopy (SEM) images and energy dispersive X-ray spectra (EDS) analysis were performed using a PHENOM XL Scanning Electron Microscope. X-ray diffraction (XRD) analyses were also conducted on the samples treated at different temperatures using an X-ray diffractometer (Regiku Miniflex 600).

3 Results

3.1 Distribution of $CaCO_3$ and chemical efficiency

The average $CaCO_3$ content and its uniformity along a soil specimen affecting the strength of MICP-treated soil specimen, and therefore the distribution of $CaCO_3$ content along the height of the sand column were measured and reasults are shown in Fig. 3.

CaCO₃ content decreases gradually along the injection direction of BS and CS in the sand column (Fig. 3 a and b). In general, the heterogeneity in the distribution of CaCO₃ precipitation along the height of the sand column increased with the average CaCO₃ content (Fig. 3a and b). Temperature 4°C and 50°C resulted in relativley low average CaCO₃ contents, and the highest average CaCO₃ contents were obtained when temperature was 35°C, shown by the comparison between the tests where BS was injected once (Tests 1-4, Fig. 3a). Increasing BS injection numbers, increases the average CaCO₃.

Chemical conversion efficiency is calculated by utilizing the average $CaCO_3$ and thereby these two parameters are consistent. The chemical conversion efficiency of the sample at 35°C was 67%, which is significantly higher than those samples treated at other temperatures, followed by 46.4% at 20°C, and decreased significantly to 19.6% and 17.3% at 4°C and 50°C, respectively. The fitted line presents a nonlinear polynomial relationship having a strong correlation coefficient of 0.998 (Fig. 3c). At 50°C, the chemical conversion efficiency has a good logarithmic fit with the injection number of BS, and the correlation coefficient is 0.997. As the BS injection number increased from 1 to 6, the chemical conversion efficiency of the sample increased from 17.3% to 56.5% (Fig. 3d).

3.2 Stress-strain and volumetric-strain responses

Fig. 4 shows the stress-strain behavior of samples treated at temperatures 4, 20, 35 and 50°C and those at 50°C treated with different BS injection numbers. All samples showed strain softening behavior, which is consistent with the results of Feng and Montoya (2016), Xiao et al. (2019), and Gao et al. (2019) obtained at temperatures of 25 °C. The strain softening characteristics of samples treated at 20°C (Fig. 4b) and 35°C (Fig. 4c) were more pronounced since a clearer peak was observed. Brittle behavior is observed when the axial strain is in the range of 0.5%-2.2% (Fig. 4 b and c), where the sample reached its peak state. In contrast, the specimens at 4°C and 50°C showed gradual softening after reaching

the peak strength, as shown in Fig. 4 a and d. The peak strength of samples at different temperatures increased with the increase of confining pressure as expected, but the brittleness level seemed not to be affected (Fig.4 a-d).

The stress-strain curves of samples under different BS injection numbers (N) at 50°C are shown in Fig. 4 d-f. All samples exhibited strain softening with the BS injection number having a significant effect on the peak strength and brittle behavior of the sample. When the BS injection number was 3 and 6, the samples showed similar stress-strain characteristics as those treated at 20°C and 35°C, including the increase of peak stress with the increase of confining pressure and the brittle behavior when the strain was small.

Fig. 5 presents the variation of volumetric strain with axial strain behavior of samples under different confining pressures (representing various depths of the soil) at different experimental conditions. All specimens showed volumetric expansion at small strain while exhibited dilative response under large strain. The results show that the treatment temperature has a significant effect on both the volumetric behavior of the specimens at small strain and the dilatant behaviour at large strain. For example, samples treated at 4 and 50°C showed less contraction at small strain than that of 20°C and 35°C (Fig. 5 a-d). The dilatancy of the sample treated at 35° C was the largest when the strain was 15% for a given confining pressure (for example at 100 kPa), followed by 20°C, 4°C and 50°C, as shown in Fig.5 a-d. The volumetric behavior of samples is related to the treatment temperature, which directly affects the CaCO₃ content. In other words, the higher the carbonate content is, the higher dilatancy at large strain occurs at a given confining pressure. Similar results were also reported by several published studies (e.g., Lin et al. 2016; Xiao et al. 2019; Wu et al. 2021) obtained at room temperatures. Fig. 5d-f shows the relationship between volumetric strain and axial strain of samples treated at 50°C with varied injection numbers of BS. The samples treated at 50°C showed significant dilatancy improvement with the increase of BS injection numbers, and contraction at small strain can be observed for all samples.

3.3 Peak strength and residual strength

Peak strength (q_p) and residual strength (q_R) can be obtained from the stressstrain curves. The variation of q_p and q_R with temperature (T) and confining pressure $(_3)$ is presented herein. Fig.6 a and c shows the measured q_p -T- $_3$ and q_R -T- $_3$ relationship for the MICP-treated samples at different temperatures with BS injected only once (Tests 1-4). It can be observed that the peak strength and residual strength increased with the increase of confining pressure. For example, the peak strength of the sample treated at 20°C increased from 623.6 kPa to 1576.8 kPa with the increase of confining pressure from 100 kPa to 400 kPa, while the residual strength increased from 271.7 kPa to 812.2 kPa. For the specimen treated at the same temperature, the peak strength and residual strength increased first and then decreased with the temperature falling in the range between 4 and 50°C. For instance, the maximum peak strength and residual strength was 322.8 kPa and 220.7 kPa at 35°C, respectively, followed by 20°C and 4°C, and the minimum was obtained at 50°C.

Fig. 6b and d presents the $q_{\rm p}$ - N_{-3} and $q_{\rm R}$ - N_{-3} relationships for the sample treated at 50°C with BS injection numbers varying from 1 to 6 (Tests 4-6). It can be seen from Fig. 6b that there is an increase of peak strength with confining pressure and BS injection number. The residual strength showed a similar increasing trend with the variation of confining pressure and BS injection number, as shown in Fig. 6d. Obviously, increasing the BS injection number has a significant improvement in the peak strength and residual strength.

Recalling the results shown in Fig. 3, it can be observed that the effect of temperature and BS injection number on the peak strength and residual strength can be attributed to the variation of CaCO₃ content. This is illustrated in Fig. 7, which presents the q_p -T-CaCO₃ content and q_R -T-CaCO₃ content relationship at various confining pressures (₃). As the average carbonate content increases the strength of the resulting specimen is higher giving a higher peak and residual deviatoric stress in the triaxial tests. The middle temperatures (20 and 35°C) seem to be optimum for strength enhancement while injecting BS multiple times seems to also facilitate strength enhancement.

3.4 Effective strength parameters

The Mohr-Coulomb failure criterion is widely used to estimate the strength parameters in geomaterials (Wood 1990). Based on the Mohr-Coulomb failure envelope, the limit equilibrium state of soil at CD triaxial tests can be expressed as follows:

(5)

This gives

(6)

where, is the deviatoric stress at limit equilibrium state, i.e., the peak deviatoric stress; c is the effective cohesion; is the effective friction angle. As illustrated in Fig. 8, the relationship between peak deviatoric stress and confining pressure of biocemented sand at different temperatures can be obtained by the following formula:

(7)

and substituting this equation to Eq. (4) gives

(8)

hence, the Mohr-Coulomb effective parameters (c and) can be derived from Eq. (8). Table 3 summarizes the fitting parameters in Eq. (7) and the effective strength parameters calculated according to Eq. (8).

Fig. 9 shows the relationship between effective strength parameters and treatment temperatures. The effective cohesion and friction angle increased first and then decreased as the temperature rose from 4 to 50°C (Fig. 9a). The lower and higher temperatures led to smaller strength parameters. For example, the effective cohesion of the samples at 4°C and 50°C was as low as 4.6 kPa and 3.1 kPa, respectively. The maximum effective cohesion value was obtained at 35°C which was 80.9 kPa. In contrast, the effect of temperature on the effective friction angle is not obvious (Fig. 9a). In the range of 4 to 50°C, the effective friction angle of the sample varied from 26.8° to 36.6°, which was significantly lower than that of cohesion. Fig. 9b shows the variation of the effective strength parameters with the BS injection number of the samples treated at 50°C. There is a linear increase between the effective cohesion and internal friction angle and the injection number of BS, with high correlation coefficients of 0.99 and 0.98 respectively.

To discuss the effect of temperature on strength of MICP-treated sand, the strength parameters (i.e., c and) derived from CD tests are plotted with CaCO₃ content, as shown in Fig. 10. For the samples with BS injected once time, the effective cohesion increased exponentially with CaCO₃ content (R^2 =0.99), while a linear relationship between effective friction angle and CaCO₃ content was observed, with a strong correlation as the R-squared value was 0.91 (Fig. 10 a). This is consistent with the results of several previous studies (e.g., Cui et al. 2017, 2021; Choi et al. 2020) conducted at temperatures of 25°C. Strong similar correlation was also found between the two strength parameters and CaCO₃ content for the samples treated at 50°C and at various BS injection numbers (Fig. 10b).

3.5 Microstructure characterization

The variations in strength with respect to the treatment temperatures arise mainly because of the differences in carbonate precipitation rates and size and distribution of the crystals within the granular network. Fig. 11 a-d shows SEM images of MICP-treated sand at different temperatures with BS injected only once. More and more CaCO₃ crystals precipitated on the surface of sand particles or at the particle contacts as the temperature increased from 4 to 35° C, while they decreased dramatically for the sample treated at 50° C. This is consistent with the macro-scale measurement of CaCO₃ content (Fig. 3). The CaCO₃ precipitation in samples treated at 4° C was mainly small crystals scattered on the sand particle surface and at particle contacts. For the samples treated at 20° C and 35° C, mainly larger crystal clusters precipitated at the contact of sand particles. Although the CaCO₃ in samples treated at 50° C and 35° C.

Fig. 11 d-f show the SEM images of samples treated at 50°C with different BS injection number. In samples treated at high temperatures mainly crystal clusters precipitated. When the BS injection number was lower (e.g., BS=1),

the crystal clusters were randomly distributed on the sand surface and among sand particle contacts, whereas the crystals were similar in size to those observed at 4°C. The size of the crystal cluster increased significantly with the increase of BS injection number. For example, the average diameter of crystal clusters can reach 25 and 40 m for the sample treated with BS injected 3 times and 6 times, respectively.

The EDS and XRD results revealed that the main elements in $CaCO_3$ precipitation formed at different temperatures were Ca, C, O, and Si, as shown in Fig. 12a and the mineral content were calcite, vaterite and quartz (Fig. 12b). It can be seen from the XRD diffraction images that at different temperatures, rhombohedral calcite is predominant in the calcium carbonate precipitation, and vaterite is sporadically visible (Fig. 12b).

4 Discussion

4.1 Effect of temperature on urease hydrolysis rate and $CaCO_3$ precipitation

The influence of temperature on urease hydrolysis rate (ureolysis rate) results from combined effects of temperature on bacterial density, bacterial urease producing rate and pure urease activity (Wang et al. 2022). Bachmeier et al. (2002) investigated the effects of temperature on pure free enzyme activity and found that urease activity decreases with time at different temperatures, with higher temperatures reducing the activity more quickly than lower temperatures. Khan et al. (2019) found the urease production ability of bacterial strains increase with temperature until the temperature reached 35°C, after which urease activity started decreasing until 50°C. Wang et al. (2022) found that bacterial density decreases with time at different temperatures, with higher temperatures reducing the bacterial density more abruptly than lower temperatures.

The influence of temperature on $CaCO_3$ precipitation results from combined effects of temperature on urease hydrolysis rate (ureolysis rate), which affects the concentration of CO_3^{2-} in the reaction system and consequently affect the supersaturation ratio of reaction system (Wang et al. 2022). In addition, studies have shown that when the supersaturation ratio is the same, $CaCO_3$ precipitation rate is higher at higher temperatures (Kralj et al. 1990, 1994 and 1997). Therefore, combining the effects of temperature on urease hydrolysis rate, the effects of temperature on $CaCO_3$ precipitation rate is complex.

In Wang et al. (2022), the highest $CaCO_3$ content was obtained when temperature was 20°C, which however in the current study was obtained when temperature was 35°C. The difference might be due to the difference in the injection interval of CS in these two studies. In Wang et al. (2022), the injection interval was 48 hours, but in this study the injection interval was reduced to 24 hours. The 24 hours was sufficient enough for 0.5 M of CS to complete precipitation at either 20 or 35°C. Because bacterial activity reduced with time at 35°C, the

shorter injection interval maintained a relatively higher bacterial activity and higher ${\rm CaCO}_3$ transform efficiency.

In most of the published studies, MICP treatment is performed at room temperature and the injection interval is 12 or 24 h between BS and CS (Cheng et al. 2013; Mahawish et al. 2018b; Wang et al. 2021). However, the interval between BS and CS needs to be carefully considered at higher temperatures (e.g., 50°C) since the urease activity and density of bacteria exhibit dramatic decay with time. The results of this study indicate that 24 h injection interval is suitable for treatment temperatures lower than 35°C, while a small injection interval should be adopted at high-temperature conditions so that guarantees the expected treatment effect. In addition, increasing the BS injection number acted as a 'compensation' for bacteria decay caused by high temperature, maintaining a supersaturation state in solution, resulting in higher CaCO₃ content produced in samples compared to that injected BS only once (Fig. 5b). These results were consistent with the results of Wang et al. 2022.

In addition, temperature does not only affect CaCO₃ precipitation rate, but also affects CaCO₃ precipitation-dissolution-reprecipitation processes. At low temperatures (i.e., 4°C), bacteria were attached randomly either on the particle surface or at particle contacts which acted as a nucleation site for crystals to precipitate. Small and uniform crystal sizes were observed at low temperatures due to the low supersaturation, resulting in low nucleation rates and slow crystal growth rates (Sahrawat 1984; Kralj et al. 1997). For the samples treated at 20°C and 35°C, more crystals precipitated at contact points and less crystals at particle surfaces since bacterial aggregations were observed and the crystals accumulated around the bacterial aggregations. Bacterial aggregations are much better candidates for nucleation sites for carbonate crystals to generate comparted to single bacterial cells because they have higher urease activity as shown by Wang et al. (2022). Also, when the crystals are formed around bacteria aggregations, these newly formed crystals serve as new nucleation sites (Mitchell & Ferris 2006; Lioliou et al. 2007). This explains the findings shown in Fig10b-c, where crystals accumulated at particle contacts and grown into larger crystal clusters.

Although higher temperatures (50°C for example) reduced the supersaturation and more nucleation took place (Mullin 2001), the rapid deactivation of bacteria led to the termination of the MICP process and as a result the crystals' growth was inhibited to then stop completely. The high temperature also resulted in the precipitation of small crystal clusters preferentially around bacterial aggregations against single bacterial cells because the former tends to decay slower. In addition, in the case of multiple bacterial injections, the injected bacteria preferred to attach to calcite rather than quartz, while calcite is easier to serve as the nucleation site than quartz for crystals (Lioliou et al. 2007). As a result, the size of crystal clusters at particle contacts increased significantly as the BS injection number increased from 1 to 6 (Fig. 11d-f).

Since MICP is a complex process with many bio-chemical and medium factors

affecting its effectiveness and the $CaCO_3$ precipitation mechanism, in this study only the temperature varied and all other factors remained fixed in order to isolate the effects. The effect of temperature on $CaCO_3$ precipitation is explained by the variations in urease activity, which is the key factor that affects the ureolysis rate, the crystals nucleation and the growth rate.

4.2 Effect of temperature on the strength of bio cemented sand

The effect of temperature on the strength of bio cemented sand essentially depends on the content of $CaCO_3$ in the specimen and on the uniformity of the MICP products (Feng and Montoya 2016; Cui et al. 2017; Nafisi et al. 2020; Konstantinou et al. 2021). $CaCO_3$ precipitation has a bonding effect on sand cementation, leading to a substantial increase of cohesion (DeJong et al. 2010). Accordingly, the peak strength of the sample increased with the rising temperature due to the increase of $CaCO_3$ content (Fig. 9), which then decreased at the highest temperature of 50°C. For the samples with BS injected once time, the effective cohesion increased exponentially with $CaCO_3$ content ($R^2=0.99$), while a linear relationship between effective friction angle and $CaCO_3$ content was observed, with a strong correlation as the R-squared value was 0.91. This is consistent with the results of several previous studies (e.g., Cui et al. 2017, 2020; Chou et al. 2020).

The cohesion increases dramatically because of the presence of carbonate crystals at the contact points between the particles since the term defines the non-frictional part of the shear resistance of the soil. The trend can be clearly explained through the microstructure properties. At 20 and 35 °C more carbonate crystal clusters are observed at the contact points between the particles which contribute to strength enhancement. At lower temperatures these clusters are smaller in number while at 50 °C, the crystal clusters seem to decrease in size resulting to a smaller strength gain compared to the previous experiments.

The relationship between strength parameters and $CaCO_3$ content reveals that the effect of temperature on effective cohesion seems to be more significant than that of friction angle. The friction angle depends primarily on grain size distribution, angularity, and particle interlocking. The 'added' calcium carbonate crystals increase the particle interlocking (due to roughness) which in turn increase the friction angle. However, the effect of strength gain when cementation is added at the contact points (cohesion) is much more pronounced in MICP-treated specimens.

The residual state, the strength of the sample is essentially attributed to the non-broken calcium carbonate between the particles and the surface roughness of soil particles caused by $CaCO_3$ precipitation (Montoya and DeJong 2015; Gao et al. 2019). The cementation degraded dramatically after the peak state was reached especially in the stronger specimens, but the non-broken calcite still plays the role of cementation, and the roughness of sand particles may also

contribute to the residual strength (Fig. 10).

The peak and residual strengths also increased when multiple BS were injected because the calcite crystal clusters increased in size as a result of this. However, still the specimen that exhibited the higher strength was the one with one injection of BS at 35°C. This shows that the temperature is indeed a very important parameter that needs to be tuned specifically for the purposes of each MICP application. In cases where temperature cannot be controlled, a proposed solution is the alteration of the MICP recipe, and specifically the number of bacterial solution injections.

5 Conclusions

In this study, a series of consolidated drained triaxial compression tests were conducted to explore the effectiveness of MICP in treating granular media at different environmental temperatures. Very high and very low temperatures such as 50 and 4 $^{\circ}$ C were also considered in this experimental work to capture extreme climate conditions. Micro-scale investigation was also conducted to explore the pore-scale characteristics of the precipitated minerals (amount, size, distribution and form of carbonate crystals) which are responsible for the resulting engineering properties of the treated soils (strength, stiffness and permeability). Based on the test results and analysis presented in this paper, the main conclusions are drawn as follows:

- 1. CaCO₃ content of samples increased as the treatment temperature rose from 4 to 35°C, while it decreased dramatically at 50°C. During triaxial tests, all MICP-treated samples exhibited strain softening and volumetric dilatancy, whereas the value of peak and residual strength showed significant variations with treatment temperature. The maximum peak and residual strength were observed from the specimens treated at 35°C, followed by 20°C. Samples treated at lower temperatures (4°C) and higher temperatures (50°C) resulted in lower peak and residual strength. The peak and residual strength of the samples treated at 50°C were significantly improved by increasing the BS injection number.
- 2. The variation of effective strength parameters (i.e., c and) with treatment temperature was similar to that of peak strength and residual strength. With the increase of temperature, the effective cohesion and friction angle increased first and then decreased. The relationship between effective strength parameters and CaCO₃ content evidenced the difference in strength enhancement of MICP-treated sand at different temperatures can be attributed to the amount of CaCO₃ precipitation.
- 3. SEM images, EDS and XRD results demonstrated that the CaCO₃ crystals produced at different temperatures had the form of rhombohedral calcite. Most calcite crystals precipitated as clusters on particle surfaces or at particle contacts. As the temperature rose from 4 to 35°C, the crystal

clusters grew larger while they 'preferred' to precipitate at particle. At 50°C, the cluster sizes were small, and no preference in location against the sand particle surfaces over the particle contacts.

4. The mechanical behavior of MICP-treated sand is closely related to the treatment temperature that needs to be tuned specifically for the purposes of each MICP application. In cases where temperature cannot be controlled, a proposed solution is the alteration of the MICP recipe, and specifically the number of bacterial solution injections.

Acknowledgements

Y.W. acknowledges the financial support of National Natural Science Foundation of China (Grant No. 52171262, 42141003), Science and Technology Innovation Committee of Shenzhen (Grant No. JCYJ20210324103812033) and Southern Marine Science and Engineering Guangdong Laboratory (Guangzhou, Grant No. K19313901) for conducting this study.

.....

References

Al Qabany, A., Soga, K., & Santamarina, C. (2012). Factors affecting efficiency of microbially induced calcite precipitation. Journal of Geotechnical and Geoenvironmental Engineering, 138(8), 992-1001.

Al Qabany, A. & Soga, K. (2013). Effect of chemical treatment used in MICP on engineering properties of cemented soils. Géotechnique, 63, No. 4, 331-339.

ASTM (2014). Standard test method for rapid determination of carbonate content of soils. ASTM D4373-14. West Conshohocken, PA.

ASTM (2017). Standard Practice for Classification of Soils for Engineering Purposes (Unified Soil Classification System). ASTM D2487-17. West Conshohocken, PA.

ASTM (2020). Standard Test Method for Consolidated Drained Triaxial Compression Test for Soils. ASTM D7181-20. West Conshohocken, PA.

Bachmeier, K. L., Williams, A. E., Warmington, J. R., & Bang, S. S. (2002). Urease activity in microbiologically-induced calcite precipitation. Journal of Biotechnology, 93(2), 171–181.

Castro-Alonso, M. J., Montañez-Hernandez, L. E., Sanchez-Muñoz, M. A., Macias Franco, M. R., Narayanasamy, R., & Balagurusamy, N. (2019). Microbially induced calcium carbonate precipitation (MICP) and its potential in bioconcrete: microbiological and molecular concepts. Frontiers in Materials, 6, 126.

Cheng, L., & Cord-Ruwisch, R. (2012). In situ soil cementation with ureolytic bacteria by surface percolation. Ecological Engineering, 42, 64-72.

Cheng, L., Cord-Ruwisch, R., & Shahin, M. A. (2013). Cementation of sand soil by microbially induced calcite precipitation at various degrees of saturation. Canadian Geotechnical Journal, 50(1), 81-90.

Cheng, L., & Cord-Ruwisch, R. (2014). Upscaling effects of soil improvement by microbially induced calcite precipitation by surface percolation. Geomicrobiology Journal, 31(5), 396-406.

Cheng, L., & Shahin, M. A. (2016). Urease active bioslurry: a novel soil improvement approach based on microbially induced carbonate precipitation. Canadian Geotechnical Journal, 53(9), 1376-1385.

Cheng, L., Shahin, M. A., & Mujah, D. (2017). Influence of key environmental conditions on microbially induced cementation for soil stabilization. Journal of Geotechnical and Geoenvironmental Engineering, 143(1), 04016083.

Chek, A., Crowley, R., Ellis, T. N., Durnin, M., & Wingender, B. (2021). Evaluation of factors affecting erodibility improvement for MICP-treated beach sand. Journal of Geotechnical and Geoenvironmental Engineering, 147(3), 04021001.

Cheshomi, A., & Mansouri, S. (2020). Study the grain size and infiltration method effects for sand soil improvement using the microbial method. Geomicrobiology Journal, 37(4), 355-365.

Chou, Sun-Gyu, Chang, Ilhan, Lee, Minhyeong, Lee, Ju-Hyung, Han, Jin-Tae, Kwon, Tae-Hyuk. 2020. Review on geotechnical engineering properties of sands treated by microbially induced calcium carbonate precipitation (MICP) and biopolymers. Construction and Building Materials, 246: 118415.

Chow, T. T., Long, H., Mok, H.Y., Li, K.W. (2011). Estimation of soil temperature profile in Hong Kong from climatic variables. Energy and Buildings, 43(12): 3568–3575. https://doi.org/10.1016/j.enbuild.2011.09.026

Cui, M. J., Zheng, J. J., Zhang, R. J., Lai, H. J., & Zhang, J. (2017). Influence of cementation level on the strength behaviour of bio-cemented sand. Acta Geotechnica, 12(5), 971-986.

Cui, M. J., Zheng, J. J., Chu, J., Wu, C. C., & Lai, H. J. (2021). Bio-mediated calcium carbonate precipitation and its effect on the shear behaviour of calcareous sand. Acta Geotechnica, 16(5), 1377-1389.

Darby, Kathleen M., Hernandez, Gabby L., DeJong, Jason T., Boulanger, Ross W., Gomez, Michael G., Wilson, Daniel W. (2019). Centrifuge Model Testing of Liquefaction Mitigation via Microbially Induced Calcite Precipitation. Journal of Geotechnical and Geoenvironmental Engineering, 145, 10: 04019084.

DeJong, J. T., Fritzges, M. B., & Nüsslein, K. (2006). Microbially induced cementation to control sand response to undrained shear. Journal of Geotechnical and Geoenvironmental Engineering, 132(11), 1381-1392.

DeJong, J. T., Mortensen, B. M., Martinez, B. C., & Nelson, D. C. (2010). Bio-mediated soil improvement. Ecological Engineering, 36(2), 197-210. DeJong, J. T., Yafrate, N. J., & DeGroot, D. J. (2011). Evaluation of undrained shear strength using full-flow penetrometers. Journal of geotechnical and geoenvironmental engineering, 137(1), 14-26.

DeJong, J. T., Soga, K., Kavazanjian, E., Burns, S., Van Paassen, L. A., Al Qabany, & Weaver, T. (2013). Biogeochemical processes and geotechnical applications: progress, opportunities and challenges. Géotechnique, 63, No. 4, 287–301.

De Muynck, W., De Belie, N., & Verstraete, W. (2010). Microbial carbonate precipitation in construction materials: a review. Ecological engineering, 36(2), 118-136.

Dhami, N. K., Reddy, M. S., & Mukherjee, M. S. (2013). Biomineralization of calcium carbonates and their engineered applications: a review. Front Microbiol 4: 1–13.

Feng, K., & Montoya, B. M. (2016). Influence of confinement and cementation level on the behavior of microbial-induced calcite precipitated sands under monotonic drained loading. Journal of Geotechnical and Geoenvironmental Engineering, 142(1), 04015057. DOI: 10.1061/(ASCE)GT.1943-5606.0001379.

Fujita, Y., Redden, G. D., Ingram, J. C., Cortez, M. M., Ferris, F. G., & Smith, R. W. (2004). Strontium incorporation into calcite generated by bacterial ureolysis. Geochimica et cosmochimica acta, 68(15), 3261-3270.

Gao, Y., Hang, L., He, J., & Chu, J. (2019). Mechanical behaviour of biocemented sands at various treatment levels and relative densities. Acta Geotechnica, 14(3), 697-707. https://doi.org/10.1007/s11440-018-0729-3

Gao, Z., Bian, L., Hu, Y., Wang, L., & Fan, J. (2007). Determination of soil temperature in an arid region. Journal of arid environments, 71(2), 157-168.

Gomez, M. G., Anderson, C. M., Graddy, C. M., DeJong, J. T., Nelson, D. C., & Ginn, T. R. (2017). Large-scale comparison of bioaugmentation and biostimulation approaches for biocementation of sands. Journal of geotechnical and geoenvironmental engineering, 143(5), 04016124.

Gomez, M. G., Graddy, C. M., DeJong, J. T., & Nelson, D. C. (2019). Biogeochemical changes during bio-cementation mediated by stimulated and augmented ureolytic microorganisms. Scientific reports, 9(1), 1-15.

Gowthaman, Sivakumar, Iki, Takashi, Nakashima, Kazunori, Ebina, Koji, Kawasaki, Satoru. (2019). Feasibility study for slope soil stabilization by microbial induced carbonate precipitation (MICP) using indigenous bacteria isolated from cold subarctic region. SN Applied Sciences, 1 (11): 1480.

Graddy, C. M., Gomez, M. G., DeJong, J. T., & Nelson, D. C. (2021). Native bacterial community convergence in augmented and stimulated ureolytic MICP biocementation. Environmental Science & Technology, 55(15), 10784-10793.

Hu, Q., & Feng, S. (2003). A daily soil temperature dataset and soil temperature climatology of the contiguous United States. Journal of applied meteorology, 42(8), 1139-1156.

Huang, R., Huang, J. X., Zhang, C., Wen, Z. H. U. O., CHEN, Y. Y., ZHU, D. H., ... & MANSARAY, L. R. (2020). Soil temperature estimation at different depths, using remotely-sensed data. Journal of Integrative Agriculture, 19(1), 277-290.

Jiang, N. J., Soga, K., & Kuo, M. (2017). Microbially induced carbonate precipitation for seepage-induced internal erosion control in sand-clay mixtures. Journal of Geotechnical and Geoenvironmental Engineering, 143(3), 04016100.

Jiang, N. J., Liu, R., Du, Y. J., & Bi, Y. Z. (2019). Microbial induced carbonate precipitation for immobilizing Pb contaminants: Toxic effects on bacterial activity and immobilization efficiency. Science of the Total Environment, 672, 722-731.

Khan, Y. M., Munir, H., & Anwar, Z. (2019). Optimization of process variables for enhanced production of urease by indigenous Aspergillus niger strains through response surface methodology. Biocatalysis and Agricultural Biotechnology, 20 (April), 101202.

Kim, Daehyun H., Nariman Mahabadi, Jaewon Jang, and Leon A. van Paassen. 2020. Assessing the Kinetics and Pore-Scale Characteristics of Biological Calcium Carbonate Precipitation in Porous Media Using a Microfluidic Chip Experiment. Water Resources Research 56 (2). https://doi.org/10.1029/2019WR 025420.

Kralj, D., Brečević, L., & Nielsen, A. E. (1990). Vaterite growth and dissolution in aqueous solution I. Kinetics of crystal growth. Journal of Crystal Growth, 104(4), 793-800.

Kralj, D., Brečević, L., & Nielsen, A. E. (1994). Vaterite growth and dissolution in aqueous solution II. Kinetics of dissolution. Journal of crystal growth, 143(3-4), 269-276.

Kralj, D., Brečević, L., & Kontrec, J. (1997). Vaterite growth and dissolution in aqueous solution III. Kinetics of transformation. Journal of crystal growth, 177(3-4), 248-257.

Konstantinou, C., Biscontin, G., Jiang, N. J., & Soga, K. (2021). Application of microbially induced carbonate precipitation to form bio-cemented artificial sandstone. Journal of Rock Mechanics and Geotechnical Engineering, 13(3), 579-592.

Konstantinou, C., Wang, Y., Biscontin, G., & Soga, K. (2021). The role of bacterial urease activity on the uniformity of carbonate precipitation profiles of bio-treated coarse sand specimens. Scientific reports, 11(1), 1-17.

Kumarathunge, M. D. P., Waidyarathne, K. P., & Samita, S. (2013). The estimation of daily mean soil temperature at different depths in three Agroecological regions of Sri Lanka.

Lee, Min Lee, Wei Soon Ng, and Yasuo Tanaka. 2013. Stress-Deformation and Compressibility Responses of Bio-Mediated Residual Soils. Ecological Engineering 60 (November): 142–49. https://doi.org/10.1016/j.ecoleng.2013.07.034.

Li, M., Cheng, X., & Guo, H. (2013). Heavy metal removal by biomineralization of urease producing bacteria isolated from soil. International Biodeterioration & Biodegradation, 76, 81-85.

Lin, H., Suleiman, M. T., Brown, D. G., & Kavazanjian Jr, E. (2016). Mechanical behavior of sands treated by microbially induced carbonate precipitation. Journal of Geotechnical and Geoenvironmental Engineering, 142(2), 04015066.

Liu, Bo, Chao Sheng Tang, Xiao Hua Pan, Cheng Zhu, Yao Jia Cheng, Jin Jian Xu, and Bin Shi. 2021. Potential Drought Mitigation Through Microbial Induced Calcite Precipitation-MICP. Water Resources Research 57 (9). https://doi.org/10.1029/2020WR029434.

Lioliou, M. G., Paraskeva, C. A., Koutsoukos, P. G., & Payatakes, A. C. (2007). Heterogeneous nucleation and growth of calcium carbonate on calcite and quartz. Journal of colloid and interface science, 308(2), 421-428.

Mahawish, A., Bouazza, A., & Gates, W. P. (2018a). Improvement of coarse sand engineering properties by microbially induced calcite precipitation. Geomicrobiology Journal, 35(10), 887-897.

Mahawish, Aamir, Bouazza, Abdelmalek, Gates, Will P. (2018b). Effect of particle size distribution on the bio-cementation of coarse aggregates. Acta Geotechnica, 13(4): 1019-1025.

Martinez, B. C., DeJong, J. T., Ginn, T. R., Montoya, B. M., Barkouki, T. H., Hunt, C., Tanyu, B., Major, D. (2013). Experimental Optimization of Microbial-Induced Carbonate Precipitation for Soil Improvement. Journal of Geotechnical and Geoenvironmental Engineering, 139(4): 587-598.

Minto, James M., Erica MacLachlan, Gráinne el Mountassir, and Rebecca J. Lunn. 2016. Rock Fracture Grouting with Microbially Induced Carbonate Precipitation. Water Resources Research 52 (11): 8827–44. https://doi.org/10.1002/2016WR018884.

Mitchell, J. K., & Santamarina, J. C. (2005). Biological considerations in geotechnical engineering. Journal of geotechnical and geoenvironmental engineering, 131(10), 1222-1233.

Mitchell, A. C., & Ferris, F. G. (2006). The influence of Bacillus pasteurii on the nucleation and growth of calcium carbonate. Geomicrobiology Journal, 23(3-4), 213-226.

Montoya, B. M., & DeJong, J. T. (2015). Stress-strain behavior of sands cemented by microbially induced calcite precipitation. Journal of Geotechnical and Geoenvironmental Engineering, 141(6), 04015019.

Mortensen, B. M., Haber, M. J., DeJong, J. T., Caslake, L. F., & Nelson, D. C. (2011). Effects of environmental factors on microbial induced calcium carbonate precipitation. Journal of applied microbiology, 111(2), 338-349.

Mugwar, A. J., & Harbottle, M. J. (2016). Toxicity effects on metal sequestration by microbially-induced carbonate precipitation. Journal of Hazardous Materials, 314, 237-248.

Mullin, J. W. (2001). Crystallization. Elsevier.

Nafisi, A., Montoya, B. M., & Evans, T. M. (2020). Shear strength envelopes of biocemented sands with varying particle size and cementation level. Journal of Geotechnical and Geoenvironmental Engineering, 146(3), 04020002.

Naveed, Muhammad, Jiangong Duan, Shahab Uddin, Muhammad Suleman, Yang Hui, and Hongyu Li. 2020. Application of Microbially Induced Calcium Carbonate Precipitation with Urea Hydrolysis to Improve the Mechanical Properties of Soil. Ecological Engineering. Elsevier B.V. https://doi.org/10.1016/j.ecoleng.2020.105885.

Okwadha, G. D., & Li, J. (2010). Optimum conditions for microbial carbonate precipitation. Chemosphere, 81(9), 1143-1148.

Omoregie, Armstrong Ighodalo, Ghazaleh Khoshdelnezamiha, Nurnajwani Senian, Dominic Ek Leong Ong, and Peter Morin Nissom. 2017. Experimental Optimisation of Various Cultural Conditions on Urease Activity for Isolated Sporosarcina Pasteurii Strains and Evaluation of Their Biocement Potentials. Ecological Engineering 109 (December): 65–75. https://doi.org/10.1016/j.ecoleng.2017.09.012.

Peng, J., & Liu, Z. (2019). Influence of temperature on microbially induced calcium carbonate precipitation for soil treatment. Plos one, 14(6), e0218396.

Sahrawat, K. L. (1984). Effects of temperature and moisture on urease activity in semi-arid tropical soils. Plant and soil, 78(3), 401-408.

Soon, N. W., Lee, L. M., Khun, T. C., & Ling, H. S. (2014). Factors affecting improvement in engineering properties of residual soil through microbial-induced calcite precipitation. Journal of Geotechnical and Geoenvironmental Engineering, 140(5), 04014006.

Sun, X., Miao, L., Tong, T., & Wang, C. (2019). Study of the effect of temperature on microbially induced carbonate precipitation. Acta Geotechnica, 14(3), 627-638.

Tobler, Dominique J., James M. Minto, Gráinne el Mountassir, Rebecca J. Lunn, and Vernon R. Phoenix. 2018. Microscale Analysis of Fractured Rock Sealed With Microbially Induced CaCO3 Precipitation: Influence on Hydraulic

and Mechanical Performance. Water Resources Research 54 (10): 8295–8308. https://doi.org/10.1029/2018WR023032.

Torres-Aravena, Á. E., Duarte-Nass, C., Azócar, L., Mella-Herrera, R., Rivas, M., & Jeison, D. (2018). Can microbially induced calcite precipitation (MICP) through a ureolytic pathway be successfully applied for removing heavy metals from wastewaters. Crystals, 8(11), 438.

Umar, Murtala, Khairul Anuar Kassim, and Kenny Tiong Ping Chiet. 2016. Biological Process of Soil Improvement in Civil Engineering: A Review. Journal of Rock Mechanics and Geotechnical Engineering 8 (5): 767–74. https://doi.org/10.1016/j.jrmge.2016.02.004.

van Paassen, L. A., Ghose, Ranajit, van der Linden Thomas, J. M., van der Star Wouter, R. L., van Loosdrecht Mark, C. M. 2010. Quantifying Biomediated Ground Improvement by Ureolysis: Large-Scale Biogrout Experiment. Journal of Geotechnical and Geoenvironmental Engineering, 136(2): 1721-1728

Wang Yuze. (2018). Microbial-Induced Calcium Carbonate Precipitation: from Micro to Macro Scale. PhD. thesis, University of Cambridge, United Kingdom.

Wang, Y. J., Han, X. L., Jiang, N. J., Wang, J., & Feng, J. (2020). The effect of enrichment media on the stimulation of native ureolytic bacteria in calcareous sand. International Journal of Environmental Science and Technology, 17(3), 1795-1808.

Wang, Y., Soga, K., DeJong, J. T., & Kabla, A. J. (2021). Effects of Bacterial Density on Growth Rate and Characteristics of Microbial-Induced CaCO₃ Precipitates: Particle-Scale Experimental Study. Journal of Geotechnical and Geoenvironmental Engineering, 147(6), 04021036.

Wang, Y., Wang, Y., Soga, K., DeJong, J. T., & Kabla, A. J. (2022). Microscale investigations of temperature-dependent Microbial-Induced Carbonate Precipitation (MICP) in the temperature range 4-50 {\deg} C. arXiv preprint arXiv:2202.09815.

Warren, L. A., Maurice, P. A., Parmar, N., & Ferris, F. G. (2001). Microbially mediated calcium carbonate precipitation: implications for interpreting calcite precipitation and for solid-phase capture of inorganic contaminants. Geomicrobiology Journal, 18(1), 93-115.

Weinhardt, Felix, Holger Class, Samaneh Vahid Dastjerdi, Nikolaos Karadimitriou, Dongwon Lee, and Holger Steeb. 2021. Experimental Methods and Imaging for Enzymatically Induced Calcite Precipitation in a Microfluidic Cell. Water Resources Research 57 (3). https://doi.org/10.1029/2020WR029361.

Wood, D. M. (1990). Soil behaviour and critical state soil mechanics. Cambridge university press.

Wu, C., Chu, J., Cheng, L., & Wu, S. (2019). Biogrouting of aggregates using premixed injection method with or without pH adjustment. Journal of Materials

in Civil Engineering, 31(9), 06019008.

Wu, S., Li, B., & Chu, J. (2021). Stress-Dilatancy Behavior of MICP-Treated Sand. International Journal of Geomechanics, 21(3), 04020264.

Xiao, Peng, Liu, Hanlong, Xiao, Yang, Stuedlein, Armin W., Evans, T. Matthew (2018). Liquefaction resistance of bio-cemented calcareous sand. Soil Dynamics and Earthquake Engineering, 107: 9-19.

Xiao, Y., Wang, Y., Desai, C. S., Jiang, X., & Liu, H. (2019). Strength and deformation responses of biocemented sands using a temperature-controlled method. International Journal of Geomechanics, 19(11), 04019120.

Xiao, Y., Zhao, C., Sun, Y., Wang, S., Wu, H., Chen, H., & Liu, H. (2021). Compression behavior of MICP-treated sand with various gradations. Acta Geotechnica, 16(5), 1391-1400.

Zamani, A., Montoya, B. M., & Gabr, M. A. (2019). Investigating challenges of in situ delivery of microbial-induced calcium carbonate precipitation (MICP) in fine-grain sands and silty sand. Canadian Geotechnical Journal, 56(12), 1889-1900.







Figure 3 CaCO₃ distribution along sand columns and chemical transform efficiency of MICP: (a, c) specimens treated at
 different temperatures with bacteria introduced only once; (b, d) specimens treated at 50°C with bacteria introduced once,
 twice, and three times, respectively



59 Figure 4 Stress-strain behaviour for the specimen treated at: (a) T=4°C; (b) T=20°C; (c) T=35°C; (d) T=50°C (BS1); (e)

T=50°C (BS3); (f) T=50°C (BS6)

61



68 Figure 5 Relationship between axial strain and volumetric strain: (a) T=4°C; (b) T=20°C; (c) T=35°C; (d) T=50°C (BS1);

(e) T=50°C (BS3); (f) T=50°C (BS6)



Figure 6 Measured and fitted relationship of q_p -T- σ_3 and q_R -T- σ_3 : (a, c) specimens treated at different temperatures with bacteria introduced only once; (b, d) specimens treated at 50°C with bacteria introduced once, twice, and three times,

respectively



85 Figure 7 Relationships between peak (a) and residual deviatoric stress (b) with average CaCO₃ content

- -

- -













(b)



Figure 12 (a) EDS spectrum of specimens treated at different temperature; (b and c) The XRD results of
biocemented sand at different temperatures: Q indicates Quartz, Ca indicates calcite.

- -